Author's response to reviews

Title: The glutamate transport inhibitor DL-Threo-beta-Benzylxoyaspartic acid (DL-TBOA) differentially affects SN38- and oxaliplatin-induced death of drug-resistant colorectal cancer cells

Authors:

Elena Pedraz-Cuesta (elenapedraz@gmail.com)
Sandra Christensen (sandra.christensen@outlook.com)
Anders A Jensen (aa@sund.ku.dk)
Niels F Jensen (nielsfrankjensen@gmail.com)
Lennart Bunch (Lebu@sund.ku.dk)
Maria U Romer (maria.unni.romer@gmail.com)
Nils Brunner (nbr@sund.ku.dk)
Jan Stenvang (stenvang@sund.ku.dk)
Stine F Pedersen (sfpedersen@bio.ku.dk)

Version: 5
Date: 24 April 2015

Author's response to reviews:

See uploaded cover letter